Author’s response to reviews

Title: In-School adolescents' weight status and blood pressure profile in South-western Nigeria: urban-rural comparison

Authors:

Gabriel Omisore (akinlolu.omisore@uniosun.edu.ng)  
Bridget Omisore (bridgetomisore@yahoo.com)  
Emmanuel Abioye-Kuteyi (eakuteyi@gmail.com)  
Ibrahim Bello (bello.ibrahim@gmail.com)  
samuel olowookere (sanuolowookere@yahoo.com)  

Version: 3 Date: 23 Nov 2017  

Author’s response to reviews:

The Editor,  
BMC Obesity  

Dear Editor,  

Response to reviewers  

Thank you for accepting our article OBSY-D-17-00012R2 for publication in your highly esteemed journal. We have done all the reviewers requested and the manuscript is ready.  

S/No  

Reviewers comment  

Response  

1. At this stage, we ask that you submit a clean version of your manuscript and do not include track changes or highlighting.  

Done as requested
2. Please also ensure that your revised manuscript conforms to the journal style, which can be found at the Instructions for Authors on the journal homepage.

Done as requested

3. Editor Comments:

1) Title page: please order author names as first name(s) and surname.

2) Declaration section should be moved to appear after the Discussion

3) Author contributions: please use author initials rather than surname and initials, i.e. SAO

4) Introduction should be changed to Background

5) Please ask a native English speaker to proofread your manuscript to ensure that grammar/stylistic errors are correct.

6) An unmarked Word file should be provided (i.e. with track changes and highlighting removed)

Done as requested

4. Reviewer reports:

Xuefeng Liu (Reviewer 1): For tables 6 and 7, there are many redundant 1s for reference groups. You may delete those rows with 1s, indicating the reference in the variable name row. For example, sex (vs male) and Body Mass Index (vs normal).

Man Ki Kwok (Reviewer 2): The authors' responses and corresponding amendments have addressed my suggestions and comments.

We thanked the reviewers. The correction has been done as requested, see page 31/32

Thank you

Dr S.A. Olowookere
Corresponding author